Author's response to reviews

Title: Construction of the influenza A virus transmission tree in a college-based population: co-transmission and interactions between influenza A viruses

Authors:

Xu-Sheng Zhang (xu-sheng.zhang@phe.gov.uk)
Daniela De Angelis (dani@mrc-bsu.cam.ac.uk)

Version: 4 Date: 10 December 2015

Author's response to reviews:

We thank referee 1 for his further ‘Minor Essential Revision’. To avoid the confusion, the word “significant” has been removed and the sentence has been changed to

“…the mixing between students of different dormitories was not frequent in comparison to the mixing between students within each dormitory.” (page 5 lines 5-6)